Author’s response to reviews

Title: Radiomic features from MRI distinguish myxomas from myxofibrosarcomas

Authors:

Teresa Martin-Carreras (teresa.martin-carreras@uphs.upenn.edu)

Hongming Li (hongming.li@uphs.upenn.edu)

Kumarasen Cooper (kumarasen.cooper@uphs.upenn.edu)

Yong Fan (yong.fan@uphs.upenn.edu)

Ronnie Sebro (ronnie.sebro@uphs.upenn.edu)

Version: 2 Date: 04 Aug 2019

Author’s response to reviews:

Response to Reviewers:

Dear Editor,

Please see responses inline to reviewer/associate editor comments.

1. Ethics

-- Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

=> The study was reviewed and approved by the University of Pennsylvania Institutional Review Board (IRB) (IRB protocol number 823821).

2. Consent

-- If consent was not required, please provide details of why this was waived and the institution/committee that approved the waiver.

=> The need for signed informed consent was waived by the University of Pennsylvania Institutional Review Board (IRB) (IRB protocol number 823821). This was waived in part because in many cases the patients moved (patients with myxomas) or died (patients with myxofibrosarcomas) and no updated address was available for these patients.
3. Authors Contributions

-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

=> All authors read and approved the final manuscript.

4. Image Attribution

-- Please state in the cover letter whether the images depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time.

Should you require an alternative source you may wish to try Wikimedia Commons:

http://commons.wikimedia.org/wiki/Main_Page

=> The images depicted in Figure 1 are our own.

5. Funding

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

=> The funding bodies had no influence in the study design, data collection, analysis, interpretation of data nor in the writing of the manuscript.

6. Consent for Publication

-- If consent was not required, please provide details of why this was waived and the institution/committee that approved the waiver.

=> The study was reviewed and approved by the University of Pennsylvania Institutional Review Board (IRB) (IRB protocol number 823821). The IRB determined that the consent for publication was not required because the results presented do not disclose personal health information (PHI).

7. Formatting
Thank you for providing a response to the reviewers/Cover Letter. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.

Please also remove the title page and disclosures statement as they are not required at this stage in the peer review process.

=> The title page and disclosure statements have been removed.

8. Additional Files

Please also add a section “Additional files” where you list the following information about your supplementary material:

- File name
- Title of data
- Description of data

Please ensure that all additional files are explicitly referred to in the main text.

=> A section “Additional files” was added before the References.

9. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

=> A clean manuscript has been uploaded.

BMC Medical Imaging operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with
English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

------------------- Editorial Policies ------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements, this may cause a delay while this is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all manuscript submissions to BMC Medical Imaging must contain a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission Guidelines web page for information regarding the criteria for each sub-section (https://bmcmedimaging.biomedcentral.com/).

Where a mandatory Declarations section is not relevant to your study design or article type, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must confirm this under this sub-heading and also provide their reasons. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page (see links below).

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements

Further information about our editorial policies can be found at the following links:

Ethical approval and consent:

http://www.biomedcentral.com/about/editorialpolicies#Ethics

Availability of data and materials section:

http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials

Recipients of this email are registered users within the Editorial Manager database for this journal. We will keep your information on file to use in the process of submitting, evaluating and publishing a manuscript. For more information on how we use your personal details please see our privacy policy at https://www.springernature.com/production-privacy-policy. If you no longer wish to receive messages from this journal or you have questions regarding database management, please contact the Publication Office at the link below.